Author's response to reviews

Title: Primary mucinous adenocarcinoma in a defunctionalized urinary bladder: a case report

Authors:

Mary Taneous MD (mtaneous@mcg.edu)
Preetha Ramalingam MD (pramalingam@mcg.edu)
Donald G Mode MD (wewedr9@yahoo.com)
Jared G Heiner MD (jaredheiner@yahoo.com)
Martha K Terris MD (mterris@mcg.edu)
Jeffrey R Lee MD (jlee@mcg.edu)

Version: 3 Date: 13 February 2009

Author's response to reviews:

Thank you for the prompt review of MS: 1069749193247680, Primary mucinous adenocarcinoma in a defunctionalized urinary bladder: a case. The reviewers have provided valuable comments that have clearly improved the quality of the report. Specifically, Review 2 provided another case report to add to the prior literature on this disease. The discussion and references have been modified to reflect the addition of this article. The second reviewer has also requested recommendations on the monitoring of patients which have been added to the comments section. Reviewer 1 had no comments requiring changes to the manuscript.

The authors appreciate your time and effort in the evaluation of this manuscript.